## **Author's Response To Reviewer Comments**

Clo<u>s</u>e

We would like to thank the reviewers for their insight. The manuscript is now edited to fulfill the changes requested in the decision letter. We would like to thank the unmatched and continuous professional editorial support at Gigascience.

Clo<u>s</u>e